Author's response to reviews

Title: Effects of Carissa opaca fruits extracts on oxidative pulmonary damages and fibrosis in rats

Authors:

sumaira sahreen (sumairasahreen@yahoo.com)
M R Khan (mrkhangau@yahoo.com)
Rahmat Khan (rahmatgu_81@yahoo.com)

Version: 4 Date: 18 September 2013

Author's response to reviews: see over
Response to reviewer comments

Dear Editor

Refer to the manuscript titled below, we revised the comments with marks and hope that now the manuscript will be able for publication.

Regards

Dr. Rahmat

Reviewer's report
Title:
Effects of Carissa opaca fruits extracts on oxidative pulmonary damages and fibrosis in rats
Version:
3
Date:
12 August 2013
Reviewer:
Ganapasam Sudhandiran

Reviewer's report:
The authors have corrected the manuscript and incorporated the changes. I accept the manuscript, provided the following minor corrections are incorporated.

Fig. 2. Please mention magnification in the figure legends.
Ans: Added

References can be updated.
Ans: Revised

There are typographical and grammatical errors throughout the manuscript.
Revised

Level of interest:
An article whose findings are important to those with closely related research interests
Quality of written English:
Needs some language corrections before being published
Ans: Revised

Statistical review:
No, the manuscript does not need to be seen by a statistician.
Reviewer's report
Title: Effects of Carissa opaca fruits extracts on oxidative pulmonary damages and fibrosis in rats
Version: 3
Date: 12 August 2013
Reviewer: ARANGANATHAN SHANMUGANATHAN
Reviewer's report:
1. grammatical error "The present project was aim" can be written as The present project is aimed... or The aim of the present project was or The current study aimed to investigate...
   Ans: Revised
2. grammatical error - There are two ANDs in one sentence. Electrophoresis was performed for 45 min and DNA was observed under digital gel doc system and photographed.
   Ans: Revised

Please check English/Grammar with native speaker before submitting to the journal.
   Ans: Revised

Level of interest:
An article of importance in its field
Quality of written English:
Needs some language corrections before being published
Statistical review:
No, the manuscript does not need to be seen by a statistician.